Author’s response to reviews

Title: Risk of Bias Tools in Systematic Reviews of Health Interventions: An Analysis of PROSPERO-Registered Protocols

Authors:

Kelly Farrah (kfarr081@uottawa.ca)
Kelsey Young (kelsey.young@canada.ca)
Matthew Tunis (matthew.tunis@canada.ca)
Linlu Zhao (linlu.zhao@canada.ca)

Version: 2 Date: 18 Jul 2019

Author’s response to reviews:

Dear Dr. Malic,

Thank you and the reviewers for your further review and consideration of our manuscript entitled, "Risk of Bias Tools in Systematic Reviews of Health Interventions: An Analysis of PROSPERO-Registered Protocols."

We are delighted to hear that both reviewers have found our revised manuscript to be acceptable for publication in Systematic Reviews. For this re-submission, we have made editorial changes (all changes tracked) to further improve the clarity of the manuscript. We also made minor formatting edits to the labels for Figure 1.

Thank you very much for your time.

Sincerely,

Linlu Zhao, PhD
Scientific Project Coordinator
Public Health Agency of Canada